Author’s response to reviews

Title: Case Report: shock septic from multidrug resistant descending necrotizing mediastinitis - intensive multimodal approach IgM enriched immunoglobulins and Polymyxin B hemoperfusion

Authors:

Vincenzo Pota (vincenzo.pota@unicampania.it)
Maria Beatrice Passavanti (beatrice.passavanti@libero.it)
Pasquale Sansone (pasquale.sansone@unicampania.it)
Maria Caterina Pace (caterina.pace@libero.it)
Filomena Peluso (filomena.peluso@gmail.com)
Alfonso Fiorelli (alfonso.fiorelli@unicampania.it)
Caterina Aurilio (caterina.aurilio@unicampania.it)

Version: 1 Date: 05 Oct 2017

Author’s response to reviews:

Dear Reviewers,

I’ve rearranged the case report basing on the suggestion and underlines better the role of IgM and Polymyxin B hemoperfusion in multi drug resistant bacteria.

I’ve cleared all the abbreviations and I’ve also prepared a table explaining biochemical and other parameters.

Sincerely

Vincenzo Pota